Author's response to reviews

Title: In vivo thyroid vibro-acoustography: a pilot study

Authors:

Azra Alizad (alizad.azra@mayo.edu)
Matthew W Urban (urban.matthew@mayo.edu)
John C Morris (morris.john@mayo.edu)
Carl C Reading (reading.carl@mayo.edu)
Randall R Kinnick (kinnick.randall@mayo.edu)
James F Greenleaf (jfg@mayo.edu)
Mostafa Fatemi (fatemi@mayo.edu)

Version: 2 Date: 7 March 2013

Author's response to reviews: see over